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Abstract: Acid mine drainage (AMD), characterized by a high concentration of heavy metals, poses 

a threat to the ecosystem and human health. Bioelectrochemical system (BES) is a promising 

technology for the simultaneous treatment of organic wastewater and recovery of metal ions from 

AMD. Different kinds of organic wastewater usually contain different predominant organic 

chemicals. However, the effect of different energy substrates on AMD treatment and microbial 

communities of BES remains largely unknown. Here, results showed that different energy 

substrates (such as glucose, acetate, ethanol, or lactate) affected the startup, maximum voltage 

output, power density, coulombic efficiency, and microbial communities of the microbial fuel cell 

(MFC). Compared with the maximum voltage output (55 mV) obtained by glucose-fed-MFC, much 

higher maximum voltage output (187 to 212 mV) was achieved by MFCs fed individually with other 

energy substrates. Acetate-fed-MFC showed the highest power density (195.07 mW/m2), followed 

by lactate (98.63 mW/m2), ethanol (52.02 mW/m2), and glucose (3.23 mW/m2). Microbial community 

analysis indicated that the microbial communities of anodic electroactive biofilms changed with 

different energy substrates. The unclassified_f_Enterobacteriaceae（87.48%）was predominant in 

glucose-fed-MFC, while Geobacter species only accounted for 0.63%. The genera of 

Methanobrevibacter (23.70%), Burkholderia-Paraburkholderia (23.47%), and Geobacter (11.90%) were the 

major genera enriched in the ethanol-fed-MFC. Geobacter was most predominant in MFC enriched 

by lactate (45.28%) or acetate (49.72%). Results showed that the abundance of exoelectrogens 

Geobacter species correlated to electricity-generation capacities of electroactive biofilms. 

Electroactive biofilms enriched with acetate, lactate, or ethanol effectively recovered all Cu2+ ion (349 

mg/L) of simulated AMD in a cathodic chamber within 53 hours by reduction as Cu0 on the cathode. 

However, only 34.65% of the total Cu2+ ion was removed in glucose-fed-MFC by precipitation with 

anions and cations rather than Cu0 on the cathode. 
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1. Introduction 

Acid mine drainage (AMD) is one typical pollutant of water in many countries that have historic 

or current mining activities. Sulfide minerals present in mining wastes (e.g., open pits, mining waste 

rock, and tailings) are inevitably oxidized to form AMD when exposed to water, air, and 

chemolithotrophic acidophiles [1–3]. AMD is characterized by a high acidity and high concentration 

of toxic heavy metals/metalloids [2]. Thus, if it is not managed properly, AMD can undoubtedly cause 

considerable water and soil contamination, massive biodiversity loss in the aquatic ecosystem, and 

severe health impacts on nearby communities [4]. In order to achieve the long-term environmental 

sustainability regarding mining activities, effective and efficient technologies that can tackle the 

remediation of AMD are highly required. 

Alkaline neutralizing chemicals, such as limestone and slaked lime, are conventionally adopted 

to treat AMD by decreasing the extreme acidity and precipitating the dissolved various poisonous 

metals/metalloids as hydroxides [5]. Despite effective remediation, the large volumes of sludges 

resulted from precipitation containing heavy metals/metalloids, which are categorized as hazardous 

materials and need further safe disposal. Other active and passive remediation technologies, such as 

bioremediation, phytoremediation, electrodialysis, wetlands, and adsorption, are also commonly 

used to treat AMD [4]. However, all those technologies have the drawbacks of either low remediation 

efficiency or high cost. Besides, some of those technologies generally produce new wastes (e.g., 

sludge, brines, and spent media), which require further treatment. 

In fact, the high concentration of dissolved metals in AMD can be recovered by the 

bioelectrochemical system (BES) as valuable products to offset the cost of treatment. Therefore, the 

bioelectrochemical system is a promising technology for the treatment of AMD. The 

bioelectrochemical system is a special biological treatment process of sewage wastewater, which 

mainly utilizes the catalytic activity of electroactive microorganisms [6]. Under anaerobic conditions, 

electroactive microorganisms degrade organic pollutants and transmit electrons through external 

circuits to generate electricity [7]. As a new form of biomass energy utilization and pollutant removal, 

bioelectrochemical systems have received extensive attention due to their non-polluting 

characteristics [8,9]. Compared with a single strain, the electrogenic microbial consortium has many 

more advantages, such as higher electricity generation efficiency, a wider range of organic substrate, 

and higher coulombic efficiency [10,11]. Therefore, the enrichment and acclimation of electrogenic 

microbial consortium from environmental samples is a conventional and effective way to increase 

the power density of bioelectrochemical systems. Previous studies have shown that different energy 

substrates used to enrich electrogenic consortium can modulate the microbial community of 

electroactive biofilms [12,13]. However, studies focusing on the effects of typical energy substrates 

on the capacities of AMD treatment and microbial communities of BES are not available. 

The purposes of this study were to compare the impacts of four typical energy substrates on the 

performance, microbial communities, and capacities of AMD treatment of enriched electroactive 

biofilms. Here, single-chamber microbial fuel cells were inoculated with anaerobic sludge and fed 

with glucose, acetate, ethanol, or lactate, respectively, as energy substrates to enrich electroactive 

biofilms. The performance of enriched electroactive biofilms was evaluated after the maximum 

voltage output was reached. The microbial communities of enriched electroactive biofilms were 

analyzed by high throughput sequence technology. The AMD treatment capacities of enriched 

electroactive biofilms were evaluated in dual-chamber microbial fuel cells. In addition, the 

mechanism for copper removal on the surface of the cathode was explored. These results indicated 

that the effects of organic chemical (that is usually contained in organic wastewater) on the 

enrichment of electroactive biofilm should be first evaluated in order to obtain an efficiently 

simultaneous treatment of organic wastewater and AMD. 
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2. Materials and Methods 

2.1. The Configuration of Microbial Fuel Cell (MFC) Reactors 

A single-chamber MFC reactor was adopted to enrich electroactive biofilms (Figure 1A). The 

cube-shaped single-chamber MFC reactor with a cylindrical chamber (3 cm diameter × 4 cm length) 

was made of Perspex. Each MFC reactor (with a working volume 28 mL) consisted of a carbon brush 

(1.5 cm in radius × 3 cm in length) as anode and a carbon cloth with disk shape (projected surface 

area of 7.07 cm2) as a cathode. To save cost, the expensive Pt catalyst usually used to coat cathode was 

not adopted in this study [14]. The anode and cathode were connected by an external resistance of 

1000 Ω by titanium wire. In order to remove contaminants on the surface, both carbon brush and 

carbon cloth were soaked overnight in acetone, followed by washing with distilled water and baked 

in a muffle furnace at 450 °C for 30 min. The dual-chamber MFC reactor was adopted to treat the 

simulated acid mine drainage (Figure 1B). The dual-chamber MFC reactor consisted of an anode 

chamber (28 mL) and a cathode chamber (15 mL). The anodes enriched in these single-chamber MFCs 

were then used in the double-chamber MFC. The two chambers were separated by an anion exchange 

membrane (Hangzhou Grion Environmental Technology, Co., Ltd, Hangzhou, China). The cathode 

and the electroactive anode of the dual-chamber MFC reactor were connected by an external 

resistance of 10 Ω. The cathode of the dual-chamber MFC reactor was made of carbon cloth with a 

rectangle shape (2.5 cm in length × 0.9 cm in width) immersed in simulated AMD. 

 

Figure 1. The single-chamber (A) and double-chamber (B) microbial fuel cell (MFC) reactors. 

2.2. Startup and Operation of MFC 

The single-chamber MFCs were inoculated with anaerobic sludge obtained from a municipal 

wastewater treatment plant. Abiotic single-chamber MFCs without the inoculum of anaerobic sludge 

were set up. Duplicate single-chamber MFC reactors were set up for each energy substrate. The 

medium used to enrich electroactive biofilms contained 20 mM energy substrate (glucose, acetate, 

ethanol, or lactate, respectively), trace element solution (100 μL/L), and Wolfe’s vitamins (0.5 mL/L) 

in 50 mM phosphate buffer (4.56 g/L, Na2HPO4; 2.45 g/L, NaH2PO4; 0.31 g/L, NH4Cl; 0.13 g/L, KCl; 

0.02 g/L, CaCl2), as modified from previous study [15]. The trace element solution contained the 

following chemicals per liter: 3.00 g MgSO4, 0.25 g FeSO4·7H2O, 0.15 g ZnCl2, 0.60 g MnSO4·H2O, 0.01 

g H3BO3, 0.01 g CuSO4·2H2O, 0.03 g NiCl2·6H2O, 0.03 g Na2MoO4, 0.20 g CoCl2, 0.03 g Na2WO4·2H2O, 

and 0.15g KAl(SO4)2·12H2O. All the chemicals used in this study were analytic pure (Sinopharm 

Chemical Reagent Co., Ltd, Shanghai, China). These MFCs were operated in a fed-batch mode in a 

temperature-controlled incubator (30 °C). The medium was replaced once the output voltage of MFC 

declined below 20 mV. The medium used to maintain the growth of enriched bioelectroactive 

biofilms in the anode chamber of these dual-chamber MFCs was identical with that used for the 

single-chamber MFCs. The cathode chamber was fed with the simulated AMD that was diluted from 

the leachate of chalcopyrite bioleaching with acid water (pH 1.80) [16]. The simulated AMD mainly 
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contained 348.87 mg/L Cu2+, 45.06 mg/L Fe3+, and 7.03 mg/L Fe2+ with a pH value of 1.80. The Cu2+ 

and Fe3+ could be served as terminal electron acceptors. Abiotic double-chamber MFCs without the 

enriched electroactive biofilm were set up. Duplicate double-chamber MFC reactors containing the 

electroactive biofilms enriched with each of these different energy substrates were set up to treat the 

simulated AMD. 

2.3. Analysis and Calculations 

The voltage across the 1000 Ω external resistance of single-chamber MFCs was recorded every 

50 s by the data acquisition unit (ADAM-4017 Analog Input Model, Advantech Co., Ltd, Shenzhen, 

China) connected to the computer. The power density and polarization curve of single-chamber 

MFCs were analyzed and calculated, as described in a previous study [17]. The power density was 

normalized to the geometrical surface area of the anode. Coulombic efficiency of single-chamber 

MFCs was calculated according to a previous study [18]. Electrochemical impedance spectroscopy 

(EIS) was applied to determine the internal resistance of these single-chamber MFCs enriched with 

different energy substrates using a potentiostat (Gamry reference 600+ workstation, Philadelphia, 

Pennsylvania, USA). The EIS measurements were conducted using a three-electrode configuration, 

with a saturated Ag/AgCl reference electrode and the anode serving as the working electrode and 

the cathode as the counter electrode. For each experimental condition, the EIS measurement was 

conducted in the frequency range from 1000 kHz to 0.01 Hz with an AC amplitude of 5 mV and 

analyzed by the software of Zview. The concentration of Fe2+ and Fe3+ in the cathode chamber was 

determined using the phenanthroline method [19]. The concentration of Cu2+ was quantified with bis-

cyclohexanone oxalyldihydrazone (BCO) [20]. The pH value of catholyte was measured with a pH-

meter (SJ-4A, Leichi, Shanghai, China). 

Scanning electron micrograph (SEM, JSM-6490LV, JEOL, Tokyo, Japan) was adopted to observe 

the enriched electroactive biofilms and the structure of cathode surfaces. The energy dispersive X-ray 

spectrometry (EDXS; Elect super, EDAX AMETEK, Kleve, Germany) equipped for SEM was used to 

examine the morphologies and compositions of the deposits on cathode electrodes after the treatment 

of AMD. The products deposited on the cathode electrode were determined by the X-ray powder 

diffraction (XRD) (D8 Advance, Bruker Corporation, Karlsruhe, Germany), in which data were 

recorded in the 2θ range of 10 to 80 degree with a step of 0.02 degree. 

2.4. Genomic DNA Extraction and MiSeq Sequencing of Bioelectroactive Biofilms  

The electroactive biofilms enriched with different energy substrates in MFCs with stable output 

voltages were sampled to extract the total genomic DNA by the DNeasy PowerSoil DNA Isolation 

Kit (QIAGEN, Chatsworth, CA, USA). Illumina adapter sequence, together with the universal primer 

pair 515FmodF (5’-GTGYCAGCMGCCGCGGTAA-3’) and 806RmodR (5’-

GGACTACNVGGGTWTCTAAT-3’), were used to amplify the V4 region of the bacterial and archaeal 

16S rDNA genes. PCR amplification was performed on Applied Biosystems GeneAmp® 9700 thermal 

cycler (ABI Inc., Foster City, CA, USA). PCR system (25 μL) consisted of 1 μL of template DNA, 1 μL 

(10 nM) of each primer, 9.5 μL of DNase-free deionized water, and 12.5 μL of 2× Taq PCR Master Mix 

(TransGen, Beijing, China). Triplicate amplifications for each genomic DNA sample were amplified 

and blended to minimize potential biases of amplification, which were separated by agarose gel 

electrophoresis (2%, w/v) and recovered using AxyPrep DNA gel extraction kit (Axygen Scientific 

Inc., USA). The concentration of the recovered PCR products was measured using QuantiFluor™-ST 

Fluorometer (Promega Corporation, Madison, WI, USA). Sequencing libraries were prepared and 

sequenced by the Illumina MiSeq platform with the sequencing strategy PE250 (Shanghai Majorbio 

Bio-pharm Technology Co., Ltd, Shanghai, China). 

The raw data of 16S rRNA gene sequences from MiSeq sequencing was in FASTQ format. The 

Illumina adapter and other specific sequences were trimmed before the following process. Then, the 

pair-end reads with at least 10 bp overlap, and lower than 5% mismatches were merged using the 

Fast Length Adjustment of SHort reads (FLASH) software [21]. The sequences shorter than 240 bp, 

chimeric sequences, and low-quality sequences were filtered, trimmed, and removed [22]. 
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Operational taxonomic units (OTUs) were obtained based on the threshold of 97% similarity by using 

UPARSE [23]. The taxonomy of OTU representative sequences was phylogenetically assigned to 

taxonomic classifications by the Ribosomal Database Project (RDP) classifier at the threshold of 70% 

for confidence based on the Bayesian algorithm [24]. Community richness, Ace and Shannon indices, 

and Chao1 richness estimates were obtained by MOTHUR analysis [25]. 

3. Results and Discussion 

3.1. Effect of Different Energy Substrates on Single Chamber MFC Performance 

Different energy substrates (i.e., glucose, acetate, ethanol, or lactate) affected the startup, 

maximum voltage output, power density, and coulombic efficiency of single-chamber microbial fuel 

cells (Figure 2). The output voltage of the MFCs enriched with lactate as an energy substrate began 

to be detectable only 40 h after the inoculation with anaerobic sludge (Figure 2A). However, in order 

to generate a detectable output voltage, 120, 210, or 220 h was required, respectively, for the MFCs 

enriched with ethanol, acetate, or glucose. Compared with the maximum voltage output (55 mV) 

obtained by glucose-fed-MFC, much higher maximum voltage output (187 to 212mV) was achieved 

by MFCs fed individually with the other three energy substrates. Around 400 h after the initial 

inoculation, the output voltage of each MFC reached the maximum. Thereafter, the output voltage 

could rapidly increase to the maximum value immediately after the removal of planktonic 

microorganisms by replenishing with growth medium containing each energy substrate (Figure 2A). 

This rapid recovery of maximum output voltage indicated that the current was mainly generated by 

the sessile microorganisms on the surface of the anode. Acetate-fed-MFC showed the highest power 

density (195.07 mW/m2), followed by lactate (98.63 mW/m2), ethanol (52.02 mW/m2), and glucose (3.23 

mW/m2) (Figure 2B). As indicated by the polarization test, the output voltage of acetate-fed-MFC was 

much higher than those of other MFCs at different external resistance (Figure 2C). On the contrary, 

the output voltage of glucose-fed-MFC was the lowest (Figure 2C). Coulombic efficiencies of these 

MFCs were dependent on the energy substrates. The MFCs enriched with lactate had the highest 

coulombic efficiency (33.34%), followed by the MFCs enriched with ethanol (14.30%), acetate 

(12.53%), and glucose (1.98%). The lowest coulombic efficiency obtained by the glucose-fed-MFCs 

was consistent with the previous studies because the glucose is a fermentable substrate that can be 

utilized by diverse microorganisms besides the exoelectrogens enriched in the electroactive biofilms 

under the anaerobic condition [12,26]. 

The ohmic resistance and charge transfer resistances of these MFCs were obtained by 

electrochemical impedance spectroscopy (EIS) (Figure 3). As described in the previous study, the 

impedance at the high-frequency limit is the ohmic resistance, and the diameter of the semicircle is 

the charge transfer resistance [27]. The ohmic resistance of MFC containing the bioelectroactive 

biofilms enriched with glucose was 19.55 Ω. However, the MFCs containing the bioelectroactive 

biofilms enriched with acetate (2.81 Ω), lactate (3.39 Ω), and ethanol (5.31 Ω) had much lower ohmic 

resistance (Figure 3). The charge transfer resistances of the MFCs containing different bioelectroactive 

biofilms were also dependent on the energy substrate used for enrichment. The charge transfer 

resistances of the MFCs containing different bioelectroactive biofilms enriched with glucose, acetate, 

ethanol, and lactate were 33.67 Ω, 7.39 Ω, 15.00 Ω, and 15.38 Ω, respectively. Discrepancy regarding 

startup, maximum voltage output, power density, coulombic efficiency, and charge transfer 

resistances of single-chamber MFCs enriched with different energy substrate implied that the 

electroactive biofilms enriched on the surface of anode were different in terms of the microbial 

community. 
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Figure 2. The output voltage (A), power density (B), and polarization curve (C) of MFCs containing 

bioelectroactive biofilms enriched with different energy substrates. 

 

Figure 3. Electrochemical impedance spectroscopy (EIS) analysis of the MFCs containing the 

bioelectroactive biofilms enriched with different energy substrates. 

3.2. Microbial Community of Anodic Bioelectroactive Biofilms 

In contrast to the abiotic control, electroactive biofilms were enriched on the surface of the anode 

of MFCs when they reached the maximum output voltage, as revealed by SEM analysis (Figure 4). 

The existence of electroactive biofilms on the surface of anode demonstrated the importance of 

electroactive biofilms for the generation of electricity. These electroactive biofilms consisted of 

microorganisms with different cell morphologies. This indicated the diversity of electroactive biofilm 

regarding the microbial community. 
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Figure 4. SEM of the bioelectroactive biofilms enriched with different energy substrates. (A: abiotic 

control; B: glucose; C: acetate; D: ethanol; E: lactate). 

In order to investigate the microbial community of electroactive biofilms enriched by different 

energy substrates (i.e., glucose, acetate, ethanol, or lactate), approximately 32,430 to 97,701 high-

quality sequencing reads were obtained from each sample (Table 1). A total number of 710 OTU was 

detected in the inoculated anaerobic sludge (Figure 5). During the enrichment of the bioelectroactive 

biofilms process, there was a succession of microorganisms at the OTU level. After MFCs reached the 

stable maximum output voltage, there were 590,482,286 and 205 OTUs in the bioelectroactive biofilms 

enriched by acetate, lactate, ethanol, and glucose, respectively (Figure 5). Both the microbial 

abundance and microbial diversity of these electroactive biofilms enriched by different energy 

substrates were less than that of the inoculated anaerobic sludge, as indicated by the Shannon index 

and Simpson index listed in Table 1. These data indicated that the microbial abundance and microbial 

diversity of these electroactive biofilms were dependent on the energy substrate. 

Table 1. The α-diversity of enriched bioelectroactive biofilms. 

Sample Reads Sobs Shannon Simpson Ace Chao Coverage 

Inoculum 40818 710 5.2069 0.0138 732.31 735.02 0.9988 

Glucose 97701 205 0.7931 0.7633 363.79 347.38 0.9993 

Acetate 45985 590 2.9848 0.2534 670.25 672.18 0.9975 

Ethanol 43875 286 2.7412 0.1233 428.09 371.00 0.9981 

Lactate 32430 482 2.9694 0.2130 589.26 586.79 0.9963  

The most dominant phyla were Proteobacteria, Bacteroidetes, and Saccharibacteria in the inoculated 

anaerobic sludge (Figure 6). Both Proteobacteria and Bacteroidetes remained as the dominant phyla in 

these enriched electroactive biofilms. The proportion of Proteobacteria increased significantly, while 

the proportion of Bacteroidetes decreased remarkably in these electroactive biofilms (Figure 6A). 

Firmicutes was enriched as one of the dominant phyla in these electroactive biofilms. It was worth 

mentioning that Euryarchaeota was enriched in these electroactive biofilms, especially in the 

electroactive biofilms fed with ethanol as an energy substrate. 

The major classes in the electroactive biofilms were different from that of the anaerobic sludge 

(Figure 6B). Gammaproteobacteria, Deltaproteobacteria, and Betaproteobacteria were the three major 

classes within the inoculated anaerobic sludge and the electroactive biofilms enriched, respectively, 

with acetate, ethanol, or lactate. However, only Gammaproteobacteria constituted as the major class of 

the electroactive biofilms enriched with glucose (87.70%). Sphingobacteriia (25.48%) and 

norank_p_Saccharibacteria (7.66%) were two major classes that existed in the inoculated anaerobic 

sludge, both of which were shifted as minor constituents in these electroactive biofilms. 
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Figure 5. The number (A) and Venn diagram analysis (B) of the operational taxonomic unit (OTU) of 

these bioelectroactive biofilms enriched with different energy substrates. 

 

Figure 6. Microbial community of the bioelectroactive biofilms enriched with different energy 

substrates at phylum (A), class (B), and genus (C) levels. 
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The major genera in anodic electroactive biofilms were modulated by energy substrates (Figure 

6C). The unclassified_f_Enterobacteriaceae（87.48%）was predominant in the glucose-fed-MFC, while 

Geobacter species only accounted for 0.63%. The genera of Methanobrevibacter (23.70%), Burkholderia-

Paraburkholderia (23.47%), and Geobacter (11.90%) were the major genera enriched in the ethanol-fed-

MFC. Geobacter was most predominant in the MFC enriched by lactate (45.28%) or acetate (49.72%), 

which corroborated with a previous study [28]. Results showed that the abundance of classic 

exoelectrogens Geobacter species correlated to the electricity-generation capacities of electroactive 

biofilms. It is worth mentioning that the Euryarchaeota was enriched in these electroactive biofilms, 

especially in the electroactive biofilms fed with ethanol as an energy substrate (Table 2). Recent 

studies have shown that quorum sensing (QS) plays an important role in shaping the dynamics of 

microbial community structure and enhancing the electron transfer process in the anodic 

electroactive biofilms of MFCs [29,30]. 

Table 2. The ratio of archaea species in the bioelectroactive biofilms enriched with different energy 

substrates. 

Archaeal Genus 
Percentage in Anode Biofilm (%) 

Inoculum Glucose Acetate Ethanol Lactate 

Methanobacterium 0  0.01  0  0.04  0.15  

Methanobrevibacter 0  1.82  0  23.70  5.23  

Methanosaeta 0  0  0.02  0  0.05  

Methanosarcina 0  0  0.19  0  0  

Methanomassiliicoccus 0  0  0.25  0  0.02  

The expression of functional genes in either single strain or microbial consortium has been 

altered by various physicochemical parameters [16,31,32]. Therefore, it is necessary to identify and 

compare the important genes involved in the electron transfer for electricity generation of these 

electrochemical biofilms in MFCs by comparative metagenomic and transcriptomic analyses in the 

future. The extracellular polymeric substances (EPS) are important for the functional roles of single 

strain and consortium [30,33,34]. The EPS of electroactive biofilm contains proteins, glycoproteins, 

extracellular DNA, glycolipids, and humic substances [30]. Previous studies have shown that 

cytochrome proteins, pili, and nanowire in EPS are directly involved in electron transfer [30,35]. 

Characterization of the compositions and redox properties of the EPS of these enriched 

electrochemical biofilms will provide novel insights into the functional role of EPS in mediating 

electron transfer. 

3.3. Contribution of Electroactive Biofilms on Anolyte’s Chemical Oxygen Demand Removal and Catholyte’s 

Copper Recovery 

Different ratio of chemical oxygen demand (COD) was depleted in the anodic chamber for the 

electroactive biofilms enriched by glucose (51.32%), acetate (82.00%), ethanol (72.49%), or lactate 

(35.95%), respectively, in 53 h after replenishing with fresh growth medium for copper recovery in 

dual-chamber MFCs (Figure 7A). A high concentration of COD (1909 mg/L) was removed in the 

anolyte of MFC fed with glucose as an energy substrate. Considering the lowest electricity production 

in each batch, most of the COD removed in the anolyte of glucose-fed-MFC was ascribed to the 

anaerobic growth by non-electrogenic microorganisms. It is worth mentioning that the number of 

planktonic microorganisms in MFC fed with glucose was much higher than those in the MFCs fed 

with other energy substrates. 

The dual-chamber MFCs containing the electroactive biofilms enriched with acetate, ethanol, or 

lactate, respectively, could effectively recover copper from the acid mine drainage (Figure 7B). The 

copper in the catholyte of these MFCs decreased significantly after the initiation of the treatment of 

AMD. At the 39th h, no detectable copper ion was found in catholyte of MFCs containing the 

electroactive biofilms enriched with acetate or lactate. At the 43rd h, the copper ion in the catholyte 
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of MFC containing the electroactive biofilms fed by ethanol was also completely recovered. However, 

the dual-chamber MFC containing electroactive biofilms enriched with glucose was deficient in the 

recovery of copper (Figure 7B). Only part of the copper ion (34.65%) was removed at the 53rd h, with 

a high concentration of Cu2+ (228.00 mg/L) remaining in the catholyte. The high concentration of Cu2+ 

(310 mg/L) remained in the catholyte of abiotic control at the end of this experiment. Iron ions in the 

stimulated AMD were mainly Fe3+ (Figure 7C,D). The decrease of Fe3+ concentration in the catholyte 

of MFCs containing electroactive biofilms was partially ascribed to the bioelectrochemical reduction 

at the cathode to Fe2+ (Figure 7D). The decrease of iron ions in the catholyte of abiotic control probably 

resulted from the elevated pH value (Figure 7E). The pH values in catholyte of all these MFCs with 

electroactive biofilms were increased during the treatment of AMD. The increase in pH value was 

likely ascribed to the diffusion of anions from the anolyte across the anion exchange member and 

reacted with the protons in the catholyte. Therefore, the decrease of iron ions in the catholyte of MFCs 

with electroactive biofilms was also affected by the increased pH values. 

 

Figure 7. The changes in chemical oxygen demand (COD) in anode chamber (A), and cupric ion (B), 

ferric iron (C), ferrous iron (D), and pH value (E) of cathode chamber of the MFCs containing 

electroactive biofilms enriched with different energy substrates. 
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3.4. Morphologies of Electrode and XRD Analysis 

The color of cathodes of dual-chamber MFCs containing the electroactive biofilms enriched with 

acetate, ethanol, or lactate, respectively, turned from black to brown after 53 h of treatment of AMD 

(Figure 8). This phenomenon indicated the bioelectrochemical reduction of copper on the surface of 

the cathode. However, the color of the cathode of abiotic control and MFCs containing electroactive 

biofilms fed with glucose remained as black (Figure 8). 

 

Figure 8. The cathodes of the MFCs after the treatment of simulated AMD. (1: abiotic; 2: glucose; 3: 

acetate; 4: ethanol; 5: lactate). 

In order to better understand the copper recovery mechanism, the cathodes of dual-chamber MFCs 

after the treatment of AMD for 53 h were analyzed with SEM and XRD. The SEM micrographs of 

cathode surfaces of these MFCs containing the electroactive biofilms enriched with acetate, ethanol, or 

lactate were similar in terms of structure and morphology, which were different from that of the 

cathodes of abiotic control and the glucose-fed-MFCs (Figure 9). No deposit was observed on the 

cathodic surface of abiotic MFCs, which was further confirmed by the EDS analysis (Figure 9A). There 

were many thin segregates on the surface of cathodes of glucose-fed MFCs. Further, EDS analysis of the 

composition of these segregates clearly showed the characteristic peaks of Cu signals at 0.98, 8.06, and 

8.87 KeVs, which confirmed the formation of Cu products (Figure 9B). Besides the Cu, many other 

elements (i.e., P, S, Cl, Na and Ca) were detected as compared with the surface of cathodes of abiotic 

control MFCs. This indicated that part of the cupric ion was precipitated with other anions and cations 

on the surface of the cathode, which was not observed in previous studies. The EDS analysis showed 

that the deposits on the cathodic surface of MFCs containing the electroactive biofilms enriched with 

acetate, ethanol, or lactate mainly contained the element of Cu (Figure 9C–E).  

The XRD patterns of the cathodic surface of MFCs containing the electroactive biofilms enriched 

with acetate, ethanol, or lactate clearly demonstrated the metal copper (Cu0) with characteristic peaks 

at 43.3, 50.4, and 74.1 degrees in 2-Theta (Figure 10). However, these characteristic peaks for metal 

copper (Cu0) were absent for the cathode from the abiotic control MFCs and the MFCs fed with 

glucose. This further indicated that no copper was deposited on the cathodic surface of these MFCs. 

The decrease of copper in the catholyte of the abiotic control MFCs and the MFCs fed with glucose 

was probably ascribed to the precipitation with other anions or cations. 
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Figure 9. SEM pictures and EDS analysis of the cathodes of MFCs after the treatment of simulated 

AMD. (A: abiotic; B: glucose; C: acetate; D: ethanol; E: lactate). 

 

Figure 10. XRD analysis of the cathodes of MFCs after the treatment of simulated AMD (▼indicates 

characteristic peaks of Cu0). 

3.5. Comparison of This Study with Previous Studies 

The organic wastewater (individually simulated by four typical pure chemicals) and simulated 

AMD were simultaneously treated in dual-chamber MFCs in this study. The effect of different energy 

substrates on anodic electroactive biofilms enrichment, bioelectrochemical activity, microbial 

communities, and AMD treatment was compared. For the scale-up of the BESs to treat the real 

industrial AMD in mining sites, these pure organic chemicals should be replaced by the real organic 

wastewater available near the pollution site in order to greatly reduce the costs. Different sources of 

real organic wastewater usually contain different predominant organic chemicals (such as these 

typical chemicals used in this study). Therefore, it is necessary to evaluate the effects of different 
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energy substrates on anodic electroactive biofilms enrichment (both bioelectrochemical activity and 

microbial communities) and AMD treatment. 

It is a fact that various organics (either pure chemical or real organic wastewater) have been 

studied as an energy substrate for MFCs [36]. However, only some studies have focused on the 

comparison of the electrochemical performance of MFCs enriched with different organic substrates 

(Table 3). On comparing our study with these studies, we found out the following differences: (1). 

The organic substrates used were generally different in other studies; (2). The microbial community 

structure of the anodic electroactive biofilms enriched with different energy substrates was studied 

with high throughput sequencing technique in this study. However, two other studies analyzed the 

anodic electroactive biofilms with traditional culture-dependent technique or denaturing gradient 

gel electrophoresis (DGGE) [12,37]. In addition, simultaneous treatment of different organic 

wastewater and simulated AMD was analyzed in this study. 

MFCs have been adopted to recover some heavy metals from wastewater [38,39]. In order to be 

consistent with the target heavy metal in our study, the studies that focused on treatment with Cu2+ 

were selected and compared (Table 3). On comparing our study with these studies, we found out the 

following differences: (1). Only a single energy substrate was used in these studies; (2). The microbial 

community structures of the anodic electroactive biofilm in these studies were scarcely studied. There 

is a study that analyzed the microbial community structure of the anodic electroactive biofilm under 

the stress of different concentrations of Cu2+ in municipal wastewater in single-chamber MFC [14]. 

However, the simulated AMD was treated in the cathode chamber, while the electroactive biofilm was 

in the anode chamber in this study (two chambers were separated by an anion exchange membrane). 

Therefore, the stress of the electroactive biofilm should be negligible. Collectively, results obtained in 

this study are insightful for the enrichment of electroactive biofilms for AMD treatment. 

Table 3. Comparison of this study with other related studies. 

Energy Substrates 
Research Focuses Reference 

Microbial Community of 
Anodic Biofilm 

Wastewater 
(Containing Cu2+)  

acetate, lactate, ethanol, 
glucose 

Yes, high throughput 
sequencing technique. 

Yes This study 

acetate, butyrate, 
propionate, glucose 

Yes, using the culture-
dependent technique. No [12] 

acetate, ethanol, glucose No No [40] 

acetate, butyrate, glucose 
Yes, using denaturing 

gradient gel electrophoresis.  No [37] 

glucose, methanol, propyl 
alcohol 

Yes, high throughput 
sequencing technique. No [13] 

acetate, glucose, starch, 
dextran, butyrate No No [41] 

glucose, lactose, cheese No No [42] 

glucose, acetate, xylose No No [43] 

acetate No Yes [44] 

acetate No Yes [45] 

glucose No Yes [46] 

acetate No Yes [47] 

acetate No Yes [48] 

acetate No Yes [17] 

acetate No Yes [49] 

acetate No Yes [50] 

acetate No Yes [51] 

acetate 
Yes, high throughput 
sequencing technique; Yes [14] 
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anodic biofilm was directly 
exposed to Cu2+. 

4. Conclusions 

This study showed that different energy substrates affected the startup, maximum voltage 

output, power density, coulombic efficiency, ohmic resistance, and the charge transfer resistance of 

MFC. The microbial community structures of these electroactive biofilms were modulated by energy 

substrates during the enrichment. The abundance of classic exoelectrogens Geobacter species 

correlated with the electricity-generation capacities of different electroactive biofilms. Geobacter 

species constituted as the predominant components of the electroactive biofilms enriched with 

acetate, ethanol, or lactate, which existed as minor species in glucose-fed electroactive biofilms 

(0.63%). The MFCs containing the glucose-fed electroactive biofilms were deficient in the extraction 

of copper from AMD. On the contrary, the MFCs containing the electroactive biofilms enriched with 

acetate, ethanol, or lactate recovered almost all the Cu2+ from the AMD by electrochemical reduction 

as metal copper (Cu0) on the surface of the cathode. These results indicated that the effects of organic 

chemical (that is usually contained in organic wastewater) on the enrichment of electroactive biofilm 

should be first evaluated in order to obtain an efficient simultaneous treatment of organic wastewater 

and AMD. Further research works are needed to assess the technical feasibility of the 

bioelectrochemical system to treat AMD, such as scale-up the reactor and run in continuous mode. 
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